Author’s response to reviews

Title: 'Sussing that doctor out.' Experiences and perspectives of people affected by hepatitis C regarding engagement with private General Practitioners in South Australia: A qualitative study

Authors:

Jane Scarborough (jane.scarborough@flinders.edu.au)
Emma Miller (emma.miller@flinders.edu.au)
Paul Aylward (paul.aylward@flinders.edu.au)
Jaklin Eliott (jaklin.eliott@adealide.edu.au)

Version: 2 Date: 03 Nov 2017

Author’s response to reviews:

Editor Comments:

Thank you for making these quite extensive revisions which I hope you feel have improved the manuscript. Before I recommend acceptance, I have one further request, which again relates to the word calculation. You have largely removed it but it still appears in one subhead and one paragraph of text. Please could you make this final amendment and resubmit.

Response: As requested the authors have removed the word “calculation” which has been replaced in the manuscript by the word “perception”.

In addition some corrections were made to the formatting of the references added in the resubmitted manuscript and the Institutional details were updated for the first author.